(1) 64621-Answering Reviewers

To reviewer #1:

Thank you for your comments.

- “Early experience” text has been reduced.
- The recommended citation on IL-6 role has been added (reference #22)
- References about vaccination has been also added (references #89-97, 99), updating the text

To reviewer #2:

Thanks for your comments and recommendations.
We have reduced the first part of the text, and we added some text about vaccination (the first version of the manuscript was sent on Feb21st), with the adequate references. A conclusion on vaccination has been also added.

To reviewer #3:

We reduced the first pages of the text to make them less redundant. We tried to make a comprehensive review on the diverse clinical and therapeutics aspects of COVID-19 on liver transplant recipients.

-Epidemiological data have been reviewed

-Page 4. The pandemic has now affected more than 100 million people and have reached a death toll exceeding 2 millions[4,5].
New version
The pandemic has now affected more than 172 million people and have reached a death toll exceeding 3.7 millions[4,5].

-References

-Page 24 References with updated Access date

COVID-19 Dashboard by the Center for Systems Science and Engineering (CSSE) at Johns Hopkins University. Available at: https://coronavirus.jhu.edu/map.html [last accessed June 10, 2021]

-Page 26

Reference 22 about IL-6 has been added as suggested by reviewer #1

-Page 39-40

Addition of references related to vaccination (89-97, 99)